Author's response to reviews

Title: Sending money home: a mixed-methods study of remittances sent by migrant nurses in Ireland

Authors:

    niamh Humphries (nhumphries@rcsi.ie)
    Ruairi Brugha (rbrugha@rcsi.ie)
    Hannah McGee (hmcmgee@rcsi.ie)

Version: 6 Date: 28 July 2009

Author's response to reviews: see over
Re: Mixed Methods Study of Remittances

Dear Sir/Madam,

Please find attached a revised copy of the manuscript, entitled ‘Sending money home: a mixed-methods study of remittances sent by migrant nurses in Ireland’

Further to your recent request for a thorough proof read of the manuscript, the following formatting changes have been made.

- The unpublished references cited have been removed and in-text references to the unpublished material have been made instead.

- All references have been altered to accommodate the deletion of these references from the reference list.

- 2-3 typos found in-text have been corrected.

- A typo in figure 1 has also been corrected (Aust changed to Aus)

- An extra acknowledgement has also been added.

Thank you for taking the time to consider our re-submission and we look forward to hearing from you again shortly.

Yours sincerely,

Niamh Humphries